Author's response to reviews

Title: Impact Evaluation of a Healthy Lifestyle Intervention to Reduce Cardiovascular Disease Risk in Health Centers in San Jose, Costa Rica and Chiapas, Mexico

Authors:

Meredith P Fort (meredith.fort@gmail.com)
Sandra Murillo (smurillo@incap.int)
Erika Lopez (erikajudith2002@yahoo.com.mx)
Ana Laura Dengo (aldengo@earth.ac.cr)
Nadia Alvarado-Molina (nadia.alvaradomolina@gmail.com)
Indira de Beausset (indiradeb@gmail.com)
Maricruz Castro (maricruz.castro@unicach.mx)
Liz Peña (lpena@incap.int)
Manuel Ramírez-Zea (mramirez@incap.int)
Homero Martínez (homero@rand.org)

Version: 2 Date: 25 August 2014

Author's response to reviews: see over
Dear Sir or Madam:

We are resubmitting the original research article titled: “” in response to formatting changes. Please see each of the changes that we have made in order to share the article for peer review.

1) I have made sure that the abstract is 350 words long and has the correct headings (Background, Methods, Results, and Conclusion).

2) I have added line numbering and page numbers to the main text file of the manuscript.

3) There is a paragraph in the Methods section (toward the top of page 10) that explains the institutions that conducted an ethical review of the research protocol, and I have added in the study number that was assigned in the US-based institution.

4) I have added in a Conclusion heading in the article.

We look forward to your review of this article and to comments and suggestions that you may have for improving it for publication. Please do not hesitate to contact me should BMC Health Services Research require additional changes to the manuscript prior to its review.

Sincerely,

Meredith Fort, PhD, MPH
Affiliate Researcher
Institute of Nutrition of Central America and Panamá
Guatemala City, Guatemala